

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Vogel, Alexander; Wagner, Joachim

### **Working Paper**

Higher productivity in importing German manufacturing firms: self-selection, learning from importing, or both?

Working Paper Series in Economics, No. 106

### **Provided in Cooperation with:**

Institute of Economics, Leuphana Universität Lüneburg

Suggested Citation: Vogel, Alexander; Wagner, Joachim (2008): Higher productivity in importing German manufacturing firms: self-selection, learning from importing, or both?, Working Paper Series in Economics, No. 106, Leuphana Universität Lüneburg, Institut für Volkswirtschaftslehre, Lüneburg

This Version is available at: https://hdl.handle.net/10419/28224

### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# NORKING

# Higher Productivity in Importing German Manufacturing Firms:

Self-selection, Learning from Importing, or Both?

by Alexander Vogel and Joachim Wagner

University of Lüneburg Working Paper Series in Economics

No. 106

November 2008

www.leuphana.de/vwl/papers ISSN 1860 - 5508

# **Higher Productivity in Importing German Manufacturing Firms:**

# **Self-selection, Learning from Importing, or Both?**

# **Alexander Vogel and Joachim Wagner**

[This version: November 18, 2008]

### Abstract:

This paper uses a newly available comprehensive panel data set for manufacturing enterprises from 2001 to 2005 to document the first empirical results on the relationship between imports and productivity for Germany, a leading actor on the world market for goods. Furthermore, for the first time the direction of causality in this relationship is investigated systematically by testing for self-selection of more productive firms into importing, and for productivity-enhancing effects of imports ('learning-by-importing'). We find a positive link between importing and productivity. From an empirical model with fixed enterprise effects that controls for firm size, industry, and unobservable firm heterogeneity we see that the premia for trading internationally are about the same in West and East Germany. Compared to firms that do not trade at all two-way traders do have the highest premia, followed by firms that only export, while firms that only import have the smallest estimated premia. We find evidence for a positive impact of productivity on importing, pointing to self-selection of more productive enterprises into imports, but no evidence for positive effects of importing on productivity due to learning-by-importing.

Keywords: imports, exports, productivity, enterprise panel data, Germany

JEL classification: F14, D21

### Acknowledgements:

All computations were performed inside the Research Data Centre of the German Federal Statistical Office (Destatis). We thank Tim Hochgürtel for running our Stata do-files and checking the voluminous outputs for any violations of privacy. Furthermore, we thank Stefan Dittrich from Destatis for his invaluable help with the data from the German Turnover Tax Statistics Panel. The enterprise data used in this study are confidential but not exclusive; see Vogel and Dittrich (2008) for details on how to access these data via the Research Data Centre. To facilitate replication and extensions the do-files used in this study are available from the first author on request.

Dipl.-Ökonom Alexander Vogel Leuphana University Lueneburg Institute of Economics PO Box 2440, 21314 Lueneburg, Germany

e-mail: avogel@leuphana.de

Prof. Dr. Joachim Wagner Leuphana University Lueneburg Institute of Economics PO Box 2440, 21314 Lueneburg, Germany

e-mail: wagner@leuphana.de

### 1. Motivation

Since the mid-1990s economists used micro data at the firm-level from many countries to uncover the role that firms play in international trade. These microeconometric studies revealed a number of stylized facts regarding differences between exporting and non-exporting firms (summarized in Bernard et al. 2007) that in turn inspired theoretical models with heterogeneous firms in open economies (see the influential contributions by Melitz 2003 and Bernard et al. 2003) instead of the representative firm models from the older literature on international economics with a focus on industries or countries. Productivity differences between exporting and non-exporting firms from the same industry play a central role in both the empirical investigations and the new theoretical models. Numerous empirical studies show that exporting firms are more productive than non-exporting firms even if observed and unobserved firm characteristics are controlled for, and that there is self-selection of the more productive firms into exporting, while empirical evidence for positive effects of exporting on productivity is scarce (for a survey of the empirical literature see Wagner 2007a).

While the causes and consequences of export and its mutual relationships with productivity (and with other firm characteristics, including firm size and growth, and wages paid) are prominent topics in the recent literature on internationally active firms, imports are seldom dealt with. A case in point is the recently published Bruegel study on the internationalisation of European firms (Mayer and Ottaviano 2007) where imports are not dealt with at all. As Bernard et al. (2007: 123) recently put it, "(t)he empirical literature on firms in international trade has been concerned almost exclusively with exporting, largely due to limitations in datasets ... . As a result, the

new theories of heterogeneous firms and trade were developed to explain facts about firm export behavior and yield few predictions (if any) for firm import behavior."

This situation, however, is changing rapidly. With new datasets that include information on imports at the firm level becoming available for more and more countries a new literature (reviewed in section 2 below) is emerging since 2005 that has a focus on the links between productivity and imports. This paper contributes to the literature by presenting the first empirical results on the relationships between imports and productivity for Germany, a leading actor on the world market for goods. Furthermore, we look for the first time systematically at the direction of causality in this relationship by testing for self-selection of more productive firms into importing, and for productivity-enhancing effects of imports ('learning-by-importing').

The rest of the paper is organized as follows: Section 2 reviews the recent literature on imports and productivity. Section 3 introduces the newly available firm level panel data for Germany used in our empirical investigation. Section 4 reports productivity premia for firms active in international trade. Section 5 investigates whether more productive firms self-select into import activities. Section 6 reports findings on productivity-enhancing effects of imports. Section 7 concludes.

### 2. Literature review

In their comprehensive empirical study of firms in the U.S. that trade goods Bernard, Jensen and Schott (2005: 5) noted "that there is virtually no research documenting and analyzing importing firms". This is no longer the case. A number of recently published empirical studies based on data from a wide range of countries document

-

<sup>&</sup>lt;sup>1</sup> The relationship between exports and productivity in Germany is investigated in Bernard and Wagner (1997) and in Wagner (2002, 2007b).

the shares of firms that are exporters, importers, and two-way traders (that both export and import), or that sell or buy on the national market only, and they look at differences between these four types of firms. Differences in productivity and their relationship with different degrees of involvement in international trade are at the centre of these studies. As of today<sup>2</sup> we have evidence on this issue for Belgium (Muuls and Pisu 2007), Chile (Kasahara and Rodrigue 2005; Kasahara and Lapham 2008), Hungary (Halpern, Koren and Szeidl 2005; Altomonte and Békés 2008), India (Tucci 2005), Indonesia (Sjöholm 1999), Italy (Castellani, Serti and Tomasi 2008), Poland (Hagemejer and Kolasa 2008), Sweden (Andersson, Lööf and Johansson 2008), and the U.S. (Bernard et al. 2007).<sup>3</sup>

Details aside, the big picture that emerges from this literature can be sketched as follows: There is a positive link between importing and productivity at the firm level, documented by a significant productivity differential between firms that import and firms that do not trade internationally; the same holds for exporting. Two-way traders are more productive than firms that either only import, or only export, or do not trade at all. Often, two-way traders are the most productive group of firms, followed by importers and then exporters, while firms selling or buying on the national market come last.

\_

Amiti and Konings (2007) who investigate the productivity effects of tariff reductions on final goods and on imported intermediate inputs in Indonesia; and MacGarvie (2006) who, in a study on patent citations, reports differences in labour productivity between exporters and non-exporters, and non-importers and importers, for a sample of French firms.

<sup>&</sup>lt;sup>2</sup> The literature on the micro-econometrics of imports is growing rapidly. We are grateful for any hints to empirical studies not listed here.

<sup>&</sup>lt;sup>3</sup> Related papers include Tomiura (2007) who looks at productivity differentials between Japanese firms that export, invest abroad, and contract out manufacturing or processing tasks to other firms overseas;

How can this empirical regularity of a positive relationship between importing and productivity at the firm level be explained theoretically? In the literature arguments for both a positive impact of productivity on importing (henceforth, hypothesis H1) and for a positive impact of importing on productivity (henceforth, hypothesis H2) are discussed. While H1 is in accordance with self-selection of more productive firms into import markets, H2 points to productivity-enhancing effects of imports ('learning-by-importing'). Let us consider the arguments in turn.

To start with H1, Kasahara and Lapham (2008) extent the Melitz (2003) model to incorporate imported intermediate goods. In their model, the use of foreign intermediates increases a firm's productivity but, due to fixed costs of importing, only inherently highly productive firms import intermediates. Andersson, Lööf and Johansson (2008) point out that importing is associated with fixed costs that are sunk costs, because the import agreement is preceded by a search process for potential foreign suppliers, inspection of goods, negotiation, contract formulation etc.. Castellani, Serti and Tomasi (2008) argue in a similar way, adding that there are sunk costs of importing due to the learning and acquisition of customs procedures.

As regards H2, Andersson, Lööf and Johansson (2008) argue that there are strong arguments in favour of a causal effect of imports on productivity, because by importing a firm can exploit global specialization and use inputs from the forefront of knowledge and technology. They point to the literature on international technology diffusion that advances imports as an important vehicle for knowledge and technology transfer. Furthermore, importing intermediate products allows a firm to focus resources and to specialize on activities where it has particular strengths. Similarly, Castellani, Serti and Tomasi (2008) argue that importers may improve productivity by using higher quality foreign inputs or by extracting technology embodied in imported intermediates and capital goods. Altomonte and Békés (2008)

point to this 'learning' effect, in which importing firms acquire part of the technology incorporated in the imported goods; furthermore they mention a variety effect (in which the broader range of available intermediates contributes to production efficiency) and a quality effect caused by imported intermediates that might be of better quality than local ones (see also Halpern, Koren and Szeidl 2005, and Muuls and Pisu 2007). If importing increases productivity, this might lead firms to self-select into export markets and help to improve their success in these markets, which might contribute to an explanation of the empirical regularity that two-way traders are the most productive firms on average (see Andersson, Lööf and Johansson 2008).

From a theoretical point of view, therefore, the direction of causality between productivity and importing can run from either sides, or from both sides simultaneously. Only some of the studies mentioned above tackle this issue (or at least a part of it) empirically. In the earliest contribution to this literature Sjöholm (1999) reports some indications of a positive growth effect from imports for his sample of Indonesian firms, but he adds as a caveat that this result is sensitive to changes in the specification of the variables and the test equation. Altamonte and Békés (2008) find that adding a new trade activity - for example, starting to import - has a positive impact on the performance of Hungarian firms. Similarly, Kasahara and Rodrigue (2005) document that switching from being a non-importer to being an importer of foreign intermediates improves productivity in Chilean manufacturing plants, while the inherently more productive plants tend to use imported intermediates. They argue that their findings indicate that the direction of causality between productivity and import status goes both ways.

The bottom line, then, is that we have convincing empirical evidence on a positive relationship between importing and productivity at the level of the firm for a large and growing number of developed and developing countries, while research on

the direction of causality between productivity and import status is still in its infancy. Furthermore, none of the very few papers tackling the issue of direction of causality does so by applying the now standard approach used to uncover the direction of causality between productivity and exporting (detailed e. g. in the survey by Wagner 2007a).

### 3. Data

In our empirical investigation we use data from the German Turnover Tax Statistics Panel (described in detail in Vogel and Dittrich 2008). This data set is based on the yearly turnover tax statistics and includes information on more than 4.3 million enterprises from all economic sectors over the time period from 2001 to 2005. All enterprises with a turnover that exceeds a rather low threshold (17,500€ since 2003) are covered in the data.

For our study we focus on enterprises from manufacturing because import and export activities can only be identified for firms from this part of the economy. However, neither exports nor imports are directly recorded in the dataset. Regarding exports, the information about 'tax free turnover with input tax deduction' can be used as a proxy. This item contains mainly the exports of goods and some activities of minor importance like gold deliveries to central banks. In addition, exports of goods within the EU (intra-Community deliveries and other performances) are directly included in the dataset. Concerning import activities, imports from EU member states are reported under the item of 'intra-Community acquisitions'. The amount of imports from states beyond the EU is not included in the turnover tax statistics. In this case an import turnover tax is charged by the customs authorities. Nonetheless, this import

-

<sup>&</sup>lt;sup>4</sup> For further details, see Vogel and Dittrich (2008).

turnover tax is deductible as input tax and therefore reported in the dataset. With this information a dummy variable which shows whether the enterprise imports from non-EU states or not can be generated (taking the value 1 if the import turnover tax is greater than zero, and 0 if no import turnover tax is deducted as input tax). Therefore, it is possible to distinguish between four types of enterprises, namely enterprises that both import and export, that only export, that only import, and that neither export nor import.

Productivity is defined as labour productivity, computed as turnover per employee covered by social insurance, because information on the number of employees was matched to the data from the turnover statistics from the German business register, and these figures refer to employees covered by social insurance only. Therefore, we had to drop all enterprises without employees that were liable to pay social insurance. Note that we can not use more appropriate measures of productivity like value added per employee, or total factor productivity, because the information needed to compute these measures are lacking in the data. In our empirical investigation we will control for the industry an enterprise is active in by using information at the detailed 3-digit-industry level to take care of inter-industry variation in capital intensity and the degree of vertical integration. Furthermore, some enterprises reported either tiny or very huge amounts of turnover in some years, leading to tiny or very huge values of labour productivity. Due to data protection rules it is impossible for us to investigate the reasons for these implausible figures, and to discriminate between reporting errors, idiosyncratic events, or other causes. Given that outliers of this kind might influence findings from both descriptive statistics and econometric investigations, enterprises from the bottom and top one percent of the labour productivity distribution were excluded from all computations.

Our empirical study, therefore, is based on information for all German enterprises from the manufacturing sector in the period 2001 to 2005 that had a turnover that exceeded the (small) tax threshold and that had at least one employee covered by social insurance, excluding very small enterprises that are mostly sole proprietorships. Table 1 reports the share of enterprises that both import and export, that only export, that only import, and that neither export nor import in each year. Given that there are large differences in the participation in international trade between manufacturing firms from West and East Germany<sup>5</sup> results are reported for both parts of Germany separately.

### [Table 1 near here]

In West Germany about half of all enterprises participated in international trade. Among the trading enterprises, about 50 percent are two-way traders that both export and import, while the share of firms that only import is somewhat larger than the share of firms that only export. The share of firms that are active on the German market only declined between 2001 and 2005, while both the share of two-way traders and firms that only import increased, and the share of firms that only export remained the same. The picture for East Germany is different. The share of firms that do not participate in international trade is more than ten percentage points higher than in West Germany, and the share of both two-way traders and firms that only export is much lower in East Germany, while the share of firms that only import is even larger. Over time the share of all kinds of trading firms increased in East Germany.

<sup>&</sup>lt;sup>5</sup> For a discussion of the difference in export participation see Wagner (2008).

Table 2a and Table 2b report how many enterprises changed their status (neither export nor import; only export; only import; both export and import) between the first and the last year covered by our empirical investigation in West and East Germany, respectively. Among the firms that were active in both years the largest group in both parts of Germany is made of firms that did not change their status. This type of stability is most often found among two-way traders. Enterprises that were active in 2001 but not in 2005 are found among all four types. Status changes in and out of one of the four categories can be found in both parts of Germany, but note that switching from no trade to two-way trade (and vice versa) is a rare event. Interestingly, about half of all firms that were not active in 2001 in West Germany, and some 40 percent of these firms in East Germany, were trading in 2005, with 25.8 percent and 17.4 percent of these new firms being two-way traders that might be considered to be 'born globals'.

### [Table 2a and Table 2b near here]

### 4. Productivity premia for firms in international trade

As a first step in our empirical investigation we compare the four types of enterprises with respect to labour productivity. Results are fully in line with the big picture that emerges from the literature reviewed in section 2 above. For 2005, figures reported in Table 3 for the mean value of labour productivity show a positive link between importing and productivity at the firm level, documented by an unconditional productivity differential between firms that import and firms that do not trade internationally; and the same holds for exporting. Two-way traders are more productive than firms that either only import, or only export, or do not trade at all. In

both parts of Germany two-way traders are the most productive group of firms, followed by importers and then exporters, while firms selling on the national market only come last. All these results hold for 2001, too, and t-tests show that all these differences in means are statistically different from zero at an error level of 0.01 or better. Note that these statistically significant differences in mean labour productivity are of an economically relevant size if two-way traders or one-way traders are compared to firms that do not trade, and if two-way traders are compared to firms that either only import or only export.

### [Table 3 near here]

If one looks at differences in the mean value for both groups only, one focuses on just one moment of the productivity distribution. A stricter test that considers all moments is a test for stochastic dominance of the productivity distribution for one group over the productivity distribution for another group. More formally, let F and G denote the cumulative distribution functions of productivity e.g. for importers and non-traders. If F(x) - G(x) = 0 means that the two distributions do not differ, and first order stochastic dominance of F relative to G means that F(z) - G(z) must be less than or equal to zero for all values of z, with strict inequality for some z. Whether this holds or not is tested non-parametrically by adopting the Kolmogorov-Smirnov test.<sup>8</sup>

\_

<sup>&</sup>lt;sup>6</sup> To economize on space results for statistical tests of differences in means (or distributions – see below) are not reported in detailed tables but summarized in the text. Detailed results are available on request.

<sup>&</sup>lt;sup>7</sup> Note that the levels of labour productivity differ considerably if firms from West and East Germany are compared. This is one reason why all empirical investigations are carried out for the two parts of Germany separately.

<sup>&</sup>lt;sup>8</sup> This method has been used to discuss the issue of exports and productivity for the first time by Delgado, Farinas and Ruano (2002).

Here six Kolmogorov-Smirnov tests were performed, comparing the productivity distribution of neither exporting nor importing enterprises vs. only exporters, neither exporting nor importing enterprises vs. only importers, neither exporting nor importing enterprises vs. two-way traders, only exporters vs. two-way traders, only exporters vs. only importers, and only importers vs. two-way traders.

Given that enterprises from the four groups compared are from different industries with different values of average labour productivity (due to, for example, differences in capital intensity), and that trading and non-trading, exporting and importing enterprises are not evenly distributed among the different industries, we control for these inter-industry differences by not using the unconditional labour productivity. Instead, we use an index that is computed as the percentage difference of labour productivity in an enterprise compared to the average value of all enterprises from the same 3-digit industry. For both West Germany and East Germany, and for both years, the Kolmogorov-Smirnov test indicates (at an error level of 0.01 or smaller) that the distributions do differ, and that the distribution for firms that participate in international trade first-order stochastically dominates the distribution for non-traders. The hierarchy of distributions is the same as the one found for the mean values of the unconditional labour productivity.

Table 3 shows that the firms from the four groups differ in size (measured by the number of employees covered by social insurance), too. In both West and East Germany enterprises that do not participate in international trade at all are on average smaller than firms that trade. Among the trading firms those that only export are smaller than those that only import, while the two-way traders are much larger on average than enterprises from both other groups of trading firms. T-tests and K-S tests again show that all these differences are statistically different from zero at an

error level of 0.01 or smaller for the unconditional mean values of firm size and the whole distributions of the number of employees conditional on the 3-digit industries.

These descriptive findings for Germany fit into the big picture that emerges from the literature reviewed in section 2. The next step in our empirical investigation is a test for the existence or not of so-called trader premia, defined as the ceteris paribus percentage difference of labour productivity between enterprises from the four groups. This is motivated by the fact that firms with different forms of participation in international trade tend to differ in size (as demonstrated above) and might be concentrated in different industries. Therefore, for example, a positive unconditional productivity differential in favour of two-way traders compared to firms that do not trade at all comes at no (or only a small) surprise. The question is whether or not this differential exists if other factors related to productivity are controlled for. To test for these trader productivity premia log labour productivity is regressed on three dummy variables indicating whether or not an enterprise exports only, imports only, or is a two-way trader (using the enterprises that do not trade at all as the reference group). The empirical model is estimated using pooled data from the years 2001 to 2005. As control variables the number of employees and its squared value and a full set of interaction terms of dummies for each year and each 3-digit-industry are included in the model. The year-industry interaction terms control for time and industry specific effects like variations in output prices and labour costs (see Lichtenberg 1988, p. 425). The empirical model is specified as follows:

(1) In LP<sub>it</sub> = 
$$a + \beta_1$$
 Ex-only<sub>it</sub> +  $\beta_2$  Im-only<sub>it</sub> +  $\beta_3$  Im-and-Ex<sub>it</sub> + c Control<sub>it</sub> +  $e_{it}$ 

where i is the index of the enterprise, t it the index of the year, LP is labour productivity, Ex-only and Im-only are dummy variables for enterprises that only export and only import in year t, and Im-and-Ex is a dummy variable for two-way traders in t.

Control is the vector of control variables, and e is an error term. The trader premium, computed from the estimated coefficient  $\beta$  as  $100(\exp(\beta)-1)$ , shows the average percentage difference in labour productivity between an enterprise from the respective group of trading firms and the non-trading enterprises, controlling for the characteristics included in the vector Control.

To demonstrate the importance of distinguishing four different groups of firms according to their involvement in international trade instead of only looking at exporting versus non-exporting firms when productivity differences between internationally active and non-active firms are investigated a variant of the model (1) is estimated that includes a dummy variable that takes the value 1 if a firm is an exporter (and 0 otherwise), completely ignoring any import activities. This is a model that is a workhorse in the empirical literature on exports and productivity (surveyed in Wagner 2007a).

To control for unobserved plant heterogeneity due to time-invariant firm characteristics which might be correlated with the variables included in the empirical model and which might lead to a biased estimate of the trader premia, (1) is estimated using pooled data for the years 2001 to 2005 and including fixed enterprise effects, too.

Results are reported in Table 4. All estimated productivity premia for firms that engage in international trade are highly statistically significant and often large from an

\_

<sup>&</sup>lt;sup>9</sup> Note that the regression equation specified in (1) is not meant to be an empirical model to explain labour productivity at the firm level; the data set at hand here is not rich enough for such an exercise. Equation (1) is just a vehicle to test for, and estimate the size of, trader premia controlling for other firm characteristics that are in the data set. Furthermore, note that productivity differences at the firm level are notoriously difficult to explain empirically. "At the micro level, productivity remains very much a measure of our ignorance." (Bartelsman and Doms 2000, p. 586)

economic point of view. Controlling for fixed enterprise effects<sup>10</sup> reduces the estimated premia considerably, pointing to the role of unobserved heterogeneity and the importance of enterprise specific factors that are both important for productivity and correlated with international activities of firms, and that lead to biased estimates of trade premia in the pooled regressions. From the results for model 1 with fixed enterprise effects we see that the premia are about the same in West and East Germany. Two-way traders do have the highest premia, followed by firms that only export, while firms that only import have the smallest estimated premia. This hierarchy differs from the picture painted by the descriptive evidence reported in section 3 where it was found that firms that only import are more productive than firms that only export. A comparison of the exporter premia estimated in model 2 with the premia for firms that export only and firms that both export and import estimated in model 1 demonstrates that it is important to consider import activities, too, even if one is interested in the relationship between exports and productivity only. In part the exporter premium estimated in model 2 here is an importer premium.

### 5. Do more productive firms self-select into importing?

Descriptive evidence reported in section 3 and evidence from a panel-econometric study presented in section 4 show a positive relationship between importing and productivity at the firm level for West and East German manufacturing enterprises. This finding is in accordance with results for other countries reviewed above. As

<sup>&</sup>lt;sup>10</sup> Due to limitations concerning the size of main memory available on the computers in the research data centre, it was not possible to estimate the fixed effects model with all West German enterprises. Therefore, the mean number of observations, the mean coefficients, and the mean p-values of five 30% random samples are reported. A documentation of the results for the five random samples can be found in Table A.1 in the appendix.

discussed in the literature survey in section 2 one hypothesis to explain this stylized fact is that causality runs from productivity to imports, and that more productive firms self-select into import activities. To shed light on the empirical validity of the hypothesis that the more productive firms go abroad and import the pre-entry differences in productivity between import starters and non-importers are investigated next.

If more productive firms become importers then we should expect to find significant differences in productivity between future import starters and future nonstarters several years before some of them begin to import. A way to test whether today's import starters were more productive than today's non-importers several years back when all of them did not import is to select all firms that did not import between year t-3 and t-1, and compute the average difference in labour productivity in year t-3 between those firms who did import in year t and those who did not. Note that some of the firms labelled "import starters" might have imported several years earlier, stopped to import then, and started again at time t. Unfortunately, the panel used here is not long enough to identify these "re-starters". The data set we use in this empirical investigation covers the years 2001 to 2005. Therefore we can look at two cohorts of import starters - firms that start to import in 2005 (where t-3 corresponds to 2002, and t to 2005) and firms that start to import in 2004 (where t-3 is equal to 2001 and t to 2004). Furthermore, we can on the one hand compare firms that did not trade internationally at all between t-3 and t-1 and that start to import in t with firms that did not trade at all between t-3 and t, and on the other hand firms that exported but not imported between t-3 and t-1 and start to import in t with firms that exported but not imported between t-3 and t.

If one looks at differences in the mean value for both groups only, one focuses on just one moment of the productivity distribution. A stricter way that considers all

moments is to test for a difference in the distribution, and for stochastic dominance of the productivity distribution for future importers over the productivity distribution for future non-importers, and to apply the Kolmogorov-Smirnov test (discussed in more detail in section 4 above) to the data for year t-3 (using, like in section 4, an index that is computed as the percentage difference of labour productivity in an enterprise compared to the average value of all enterprises from the same 3-digit industry).

Results reported in Table 5A (for import starters in 2005) and Table 5B (for import starters in 2004) indicate self-selection of more productive (and larger) enterprises into import activities. Regardless of the start year t, the part of Germany, and the definition of starters and the reference group, on average the future importers were more productive and had a larger number of employees than the future non-importers three years before starting to import. If firms that did not trade internationally at all between t-3 and t-1 and that start to import in t are compared with firms that did not trade at all between t-3 and t, at an error level of 0.01 or less these average differences are statistically significantly different from zero according to t-tests, and the distribution of import starters stochastically dominates the distribution of non-starters. If firms that exported but not imported between t-3 and t-1 and start to import in t are compared with firms that exported but not imported between t-3 and t, the picture is different – the differences in productivity are never statistically significant at a usual error level, and the same holds for differences in the number of employees in East Germany.

Furthermore, labour productivity premia of future importers compared to future non-importers were estimated controlling for plant size and industry affiliation by estimating the empirical model

(2) In 
$$LP_{it-3} = a + \beta Import_{it} + c Control_{it-3} + e_{it}$$

where i is the index of the firm, t is the index of the year, LP is labour productivity in year t-3, Import is a dummy variable for current import status (1 if the firm imports in year t, 0 else), Control is a vector of control variables (the number of employees – also included in squares - and 3-digit industry dummies), and e is an error term. The pre-entry premium, computed from the estimated coefficient ß as 100(exp(ß)-1), shows the average percentage difference between today's importers and today's non-importers 3 years before starting to import, controlling for the characteristics included in the vector Control.

Results are reported in Table 6. In model 1 the coefficient shows the average percentage productivity difference at t-3 between import starters at t and enterprises with no international activities over the whole period (year t-3 to t). In model 2 the coefficient shows the average percentage productivity difference at t-3 between exporters that start to import at t and exporters that do not start to import. All point estimates are positive, and larger for East than for West Germany. In both parts of Germany the pre-entry productivity premia of import starters are statistically different from zero at a usual error level, and large from an economic point of view, when non-traders that start to import in t are compared to firms that do not trade at all over the whole period. For exporters that start to import in t compared to exporters that do not import over the whole period this is only the case in West Germany. Note, however, that the number of import starters from this group is small in East Germany (65 and 87 firms in the starter cohort 2004 and 2005, respectively; see Table 5A and 5B), and this may cause an imprecisely estimated regression coefficient.

### [Table 6 near here]

The bottom line, then, is that for German manufacturing firms we find evidence in favour of H1 - a positive impact of productivity on importing.

### 6. Do import starters become more productive?

The second hypothesis why importers can be expected to be more productive than their counterparts that buy intermediate inputs on the domestic market only points to the role of learning-by-importing. As is argued in section 2, an importing firm can exploit global specialization and use inputs from the forefront of knowledge and technology. Imports, therefore, can act as an important vehicle for knowledge and technology transfer. Furthermore, importing intermediate products allows a firm to focus resources and to specialize on activities where it has particular strengths. Knowledge flows from international sellers and competitors help to improve the postentry performance of import starters.

If importing improves productivity then we should expect to find significant differences in the rate of growth of labour productivity between import starters and firms that continue to buy intermediate inputs on the national market only during the years after the start. This hypothesis is tested by looking at the growth rate of labour productivity over the period 2004 to 2005 for a cohort of import starters in 2003 compared to the growth performance of non-importers over the same period. Furthermore, for the period 2004 to 2005 the performance of exporters that start to import in 2003 is compared to the performance of firms that only export between 2001 and 2005.

Results are reported in Table 7. On average, the productivity growth performance of import starters from both groups was better compared to non-importers in West Germany, and the same holds for the growth of the number of

employees. The big picture is the same for East Germany except for productivity growth in import starters compared to non-trading firms. All these post-entry performance differences, however, are never statistically different from zero at an error level of five percent using t-tests or Kolmogorov-Smirnov tests.

### [Table 7 near here]

Furthermore, differences in productivity growth between import starters and non-importers are investigated based on the empirical model

(3) In LP<sub>it+2</sub> - In LP<sub>it+1</sub> = 
$$a + \beta$$
 Start<sub>it</sub> +  $c$  Control<sub>it</sub> +  $e_{it}$ 

where i is the index of the firm, t is the index of the year, LP is labour productivity, Start is a dummy variable for import starters (1 if the firm starts to import in year t, 0 else), Control is a vector of control variables (the number of employees – also included in squares - and 3-digit industry dummies), and e is an error term. Results are reported in Table 8. To facilitate interpretation, the estimated coefficient for the starter-dummy has been transformed by 100(exp(ß)-1). In model 1 the transformed coefficient shows the average productivity growth premium of import starters in 2003 compared to enterprises with no international activities two years after starting to import. In model 2 the transformed coefficient shows the average productivity growth premium of exporters that start to import in 2003 compared to enterprises that only export over the whole period two years after starting to import. While the point estimates of three out of four regression coefficients are positive, none is statistically different from zero at a conventional level of significance. Therefore, again we have no evidence for learning-by-importing. Note, however, that

the number of import starters is small in East Germany (see Table 7), and this may cause imprecisely estimated regression coefficients.

### [Table 8 near here]

In line with a recent development in the literature on the impact of exporting on productivity an alternative approach to test for productivity enhancing effects of starting to import is applied next. To motivate this approach, consider the following situation: Assume that a study reports that plants entering the import market have substantially faster productivity growth in the following years than firms that keep buying intermediate inputs on the domestic market only. Does this point to a causal effect of starting to import on productivity? The answer is, obviously, no: If better firms self-select into import-starting, and if, therefore, today's import starters are 'better' than today's non-importers (and have been so in the recent past), we would expect that they should, on average, perform better in the future even if they do not start to import today. However, we cannot observe whether they would really do so because they do start to import today; we simply have no data for the counterfactual situation. So how can we be sure that the better performance of starters compared to non-importers is caused by importing (or not)? This closely resembles a situation familiar from the evaluation of active labour market programs (or any other form of treatment of units): If participants, or treated units, are not selected randomly from a population but are selected or self-select according to certain criteria, the effect of a treatment cannot be evaluated by comparing the average performance of the treated and the non-treated. However, given that each unit (plant, or person, etc.) either participated or not, we have no information about its performance in the counterfactual situation. A way out is to construct a control group in such a way that

every treated unit is matched to an untreated unit that has been as similar as possible (ideally, identical) at the time before the treatment. Differences between the two groups (the treated, and the matched non-treated) after the treatment can then be attributed to the treatment (for a comprehensive discussion, see Heckman, LaLonde and Smith 1999).

The use of a matching approach to search for effects of starting to export on productivity (and other dimensions of firm performance) has been pioneered by Wagner (2002), and it has been used in a growing number of empirical studies (surveyed in Wagner 2007a) ever since. Here, import starters in 2003 were matched with "twins" from the large group on non-importers based on characteristics of the enterprises in 2002 (the year before the starters start), 11 and the difference in the average rate of growth of labour productivity over the period 2004 to 2005 between import starters and matched non-importers is computed. This difference is the so-called average treatment effect on the treated, or ATT, the estimated effect of import start on the growth of labour productivity (see Wagner (2002) for a discussion of this method).

\_

<sup>11</sup> Matching was done by nearest neighbours propensity score matching. The propensity score was estimated from a probit regression of a dummy variable indicating whether or not a firm is an import starter in 2003 on the log of labour productivity, number of employees, and 3-digit industry dummy variables (all measured in year 2002) plus the rate of growth of labour productivity in the years 2001 to 2002. The balancing property (that requires an absence of statistically significant differences between the treatment group and the control group in the covariates after matching) is satisfied. The difference in means of the variables used to compute the propensity score were never statistically significant between the starters and the matched non-starters. The common support condition (that requires that the propensity score of a treated observation is neither higher than the maximum nor less than the minimum propensity score of the controls) was imposed by dropping import starters (treated observations) whose propensity score is higher than the maximum or lower than the minimum propensity score of the non-importers (the controls). Matching was done using Stata 10 and the psmatch2 command (version 3.0.0), see Leuven and Sianesi (2003). The results of the probit estimates used in the matching are available on request.

Results are reported in Table 9. The big picture arising from comparing import starters with matched non-importers is the same as the one sketched above based on the comparison of import starters and all non-importers. The estimated ATT is positive for three out of four cases, but it is statistically significantly different from zero (and negative) for East Germany only when non-trading enterprises that start to import in 2003 are compared to matched enterprises that do not trade at all. Therefore, from the matching approach we have no evidence in favour of the learning-by-importing hypothesis for German manufacturing enterprises.

### [Table 9 near here]

# 7. Concluding remarks

This paper uses a newly available comprehensive panel data set for manufacturing enterprises from 2001 to 2005 to present the first empirical results on the relationship between imports and productivity for Germany, a leading actor on the world market for goods. Furthermore, for the first time the direction of causality in this relationship is investigated systematically by testing for self-selection of more productive firms into importing, and for productivity-enhancing effects of imports ('learning-by-importing').

Descriptive statistics show a positive link between importing and productivity at the firm level, documented by an unconditional productivity differential between firms that import and firms that do not trade internationally; and the same holds for exporting. From an empirical model with fixed enterprise effects that controls for firm size, industry, and unobservable firm heterogeneity we see that the premia for trading internationally are about the same in West and East Germany. Two-way traders do

have the highest premia, followed by firms that only export, while firms that only import have the smallest estimated premia. We find evidence for a positive impact of productivity on importing, pointing to self-selection of more productive enterprises into imports, but no evidence for positive effects of importing on productivity due to learning-by-importing.

The empirical evidence on a positive relationship between importing and productivity at the level of the firm is in accordance with findings for a large and growing number of developed and developing countries. Research on the direction of causality between productivity and import status, however, is still in its infancy. No other of the very few papers tackling the issue of direction of causality known to us does so by applying the approach used here. Future research will hopefully show whether the lack of evidence for learning-by-importing (that is matched by a similar lack of evidence regarding learning-by-exporting, see Wagner 2007b) found for Germany is special, or whether it can be found in other developed and developing countries, too. Stylized facts based on comparable studies using data from many countries can then be used as an input for both appropriate theoretical models of heterogeneous firms that trade, and the discussion of policy conclusions based thereon.

### References

- Altomonte, Carlo and Gabor Békés (2008). 'Trading Activities, Firms and Productivity'. Bocconi University, Milan, and Hungarian Academy of Science, Budapest, mimeo, June.
- Amiti, Mary and Jozef Konings (2007). 'Trade Liberalization, Intermediate Inputs, and Productivity: Evidence from Indonesia'. *American Economic Review*, 97(5): 1611-1638.
- Andersson, Martin, Hans Lööf and Sara Johansson (2008). 'Productivity and International Trade Firm-level Evidence from a Small Open Economy', CESIS Royal Institute of Technology, Stockholm, mimeo (forthcoming, Review of World Economics).
- Bartelsman, Eric J. and Mark Doms (2000). 'Understanding Productivity: Lessons from Longitudinal Micro Data.' *Journal of Economic Literature* XXXVIII(3): 569-594.
- Bernard, Andrew B., Jonathan Eaton, J. Bradford Jensen and Samuel S. Kortum (2003). 'Plants and Productivity in International Trade'. *American Economic Review*, 93(4): 1268-1290.
- Bernard, Andrew B., J. Bradford Jensen, Stephen J. Redding and Peter K. Schott (2007). 'Firms in International Trade'. *Journal of Economic Perspectives*, 21(3): 105-130.
- Bernard, Andrew B., J. Bradford Jensen and Peter K. Schott (2005). 'Importers, Exporters, and Multinationals: A Portrait of Firms in the U.S. That Trade Goods'. NBER Working Paper Series No. 11404, June.
- Bernard, Andrew B. and Joachim Wagner (1997). 'Exports and Success in German Manufacturing'. *Review of World Economics*, 133(1): 134-157.

- Castellani, Davide, Francesco Serti and Chiara Tomasi (2008). 'Firms in International Trade: Importers and Exporters Heterogeneity in the Italian Manufacturing Industry'. University of Perugia, draft mimeo, January.
- Delgado, Miguel A., Jose C. Farinas and Sonia Ruano (2002). 'Firm productivity and export markets: a non-parametric approach.' *Journal of International Economics* 57(2): 397-422.
- Hagemejer, Jan and Marcin Kolasa (2008). 'Internationalization and Economic Performance of Enterprises: Evidence from Firm-Level Data'. Munich Personal RePEc Archive Paper No. 8720, May.
- Halpern, László, Koren Miklós and Ádám Szeidl (2005). 'Imports and Productivity'.

  Institute of Economics, Hungarian Academy of Sciences, mimeo, June.
- Heckman, James J., R. J. LaLonde and J. A. Smith. (1999). 'The economics and econometrics of active labor market programs.' In: Ashenfelter, Orley. C. and David Card (Eds.), *Handbook of Labor Economics, Vol. 3A*, Amsterdam: North-Holland, 1865-2097.
- Kasahara, Hiroyuki and Beverly Lapham (2008). 'Productivity and the Decision to Import and Export: Theory and Evidence'. CESifo Working Paper No. 2240, March (forthcoming, *Journal of International Economics*).
- Kasahara, Hiroyuki and Joel Rodrigue (2005). 'Does the Use of Imported Intermediates Increase Productivity? Plant-Level Evidence'. University of Western Ontario EPRI Working Paper Series No. 2005-7, September.
- Leuven, Edwin and Barbara Sianesi (2003). 'PSMATCH2: Stata module to perform full Mahalanobis and propensity score matching, common support graphing, and covariate imbalance testing',
  - http://ideas.repec.org/c/boc/bocode/s432001.html.

- Lichtenberg, Frank R. (1988). 'Estimation of the Internal Adjustment Costs Model

  Using Longitudinal Establishment Data.' *Review of Economics and Statistics*70(3): 421-430.
- MacGarvie, Megan (2006). 'Do Firms Learn From International Trade?' *Review of Economics and Statistics*, 88(1): 46-60.
- Mayer, Thierry and Gianmarco I. P. Ottaviano (2007). 'The Happy Few: The Internationalisation of European Firms. New Facts based on Firm-level Evidence'. Bruegel Blueprint Series Volume III, November.
- Melitz, Marc J. (2003). 'The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity'. *Econometrica*, 71(6): 1695-1725.
- Muuls, Mirabelle and Mauro Pisu (2007). 'Imports and Exports at the Level of the Firm: Evidence from Belgium'. London School of Economics and Political Science CEP Discussion Paper No. 801, June.
- Sjöholm, Frederik (1999). 'Exports, Imports and Productivity: Results from Indonesian Establishment Data'. *World Development*, 27(4): 705-715.
- Tomiura, Eiichi (2007). 'Foreign Outsourcing, exporting, and FDI: A Productivity

  Comparison at the Firm Level'. *Journal of International Economics*, 72(1):

  113-127.
- Tucci, Alessandra (2005). 'Trade, Foreign Networks and Performance: A Firm-level Analysis for India'. Centro Studi Luca D'Agliano Development Studies Working Papers No. 199, March.
- Vogel, Alexander and Stefan Dittrich (2008). 'The German Turnover Tax Statistics Panel'. Schmollers Jahrbuch / Journal of Applied Social Science Studies, 128(4): 661-670.

- Wagner, Joachim (2002). 'The Causal Effects of Exports on Firm Size and Labor Productivity: First Evidence from a Matching Approach'. *Economics Letters*, 77(2): 287-292.
- Wagner, Joachim (2007a). 'Exports and Productivity: A Survey of the Evidence from Firm Level Data'. *The World Economy*, 30(1): 60-82.
- Wagner, Joachim (2007b). 'Exports and Productivity in Germany'. *Applied Economics Quarterly*, 53(4): 353-373.
- Wagner, Joachim (2008). 'A Note on Why more West than East German Firms Export'. *International Economics and Economic Policy*, 5(4), in press.

TABLE 1

IMPORT AND EXPORT PARTICIPATION OF MANUFACTURING ENTERPRISES IN WEST AND EAST GERMANY

	neither export nor import	only export	only import	both export and import (two-way traders)	Number of total observations					
	West Germany									
2001	54.35	10.08	11.26	24.31	135,827					
2002	54.07	10.10	11.19	24.64	131,941					
2003	52.73	10.18	11.48	25.62	134,288					
2004	51.08	10.17	11.91	26.84	132,305					
2005	49.97	10.07	12.47	27.49	131,170					
		East	Germany							
2001	68.37	5.75	13.03	12.85	30,630					
2002	67.43	6.02	13.05	13.51	29,490					
2003	66.14	6.00	13.43	14.43	28,718					
2004	63.25	6.64	14.12	15.99	27,894					
2005	61.89	7.03	14.17	16.91	27,451					

Only enterprises with one or more employees liable for paying social insurance and a turnover higher than €17,081 in 2001 prices are considered. Tax groups and enterprises with a foreign legal form are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005.

TABLE 2A

TRANSITION MATRIX OF MANUFACTURING ENTERPRISES IN WEST GERMANY 2001/2005

		Enterprise status in 2005						
		Not active in 2005	Neither export nor import	Only export	Only import	Both export and import (two-way traders)	Total	
	Not active in 2001	-	19,826 (30.2) [51.5]	3,749 (28.4) [9.7]	4,958 (30.3) [12.9]	9,941 (27.6) [25.8]	38,474 (22.1) [100.0]	
	Neither export nor import	25,289 (58.6) [34.3]	39,918 (60.9) [54.1]	2,909 (22.0) [3.9]	4,375 (26.8) [5.9]	1,333 (3.7) [1.8]	73,824 (42.4) [100.0]	
Entorpriso	Only export	4,284 (9.9) [31.3]	2,128 (3.2) [15.5]	4,411 (33.4) [32.2]	493 (3.0) [3.6]	2,377 (6.6) [17.4]	13,693 (7.9) [100.0]	
Enterprise status in 2001	Only import	4,650 (10.8) [30.4]	3,048 (4.7) [19.9]	492 (3.7) [3.2]	5,134 (31.4) [33.6]	1,965 (5.4) [12.9]	15,289 (8.8) [100.0]	
	Both export and import (two-way trader)	8,908 (20.7) [27.0]	626 (1.0) [1.9]	1,651 (12.5) [5.0]	1,394 (8.5) [4.2]	20,442 (56.7) [61.9]	33,021 (18.9) [100.0]	
	Total	43,131 (100.0) [24.7]	65,546 (100.0) [37.6]	13,212 (100.0) [7.6]	16,354 (100.0) [9.4]	36,058 (100.0) [20.7]	174,301 (100.0) [100.0]	

Reported are the number of cases, the column percentages in parenthesis ( ), and the row percentages in brackets [ ]. Only enterprises with one or more employees liable for paying social insurance and a turnover higher than  $\in$ 17,081 in 2001 prices are considered. Tax groups and enterprises with a foreign legal form are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005.

TABLE 2B

TRANSITION MATRIX OF MANUFACTURING ENTERPRISES IN EAST GERMANY 2001/2005

				Enterprise sta	atus in 2005		
		Not active in 2005	Neither export nor import	Only export	Only import	Both export and import (two-way traders)	Total
	Not active in 2001	-	4,859 (28.6) [61.4]	557 (28.9) [7.0]	1,127 (29.0) [14.2]	1,377 (29.7) [17.4]	7,920 (20.5) [100.0]
	Neither export nor import	7,991 (72.0) [38.2]	10,880 (64.0) [52.0]	586 (30.4) [2.8]	1,185 (30.5) [5.7]	300 (6.5) [1.4]	20,942 (54.3) [100.0]
Enterprise	Only export	572 (5.2) [32.5]	335 (2.0) [19.0]	480 (24.9) [27.3]	88 (2.3) [5.0]	286 (6.2) [16.2]	1,761 (4.6) [100.0]
status in 2001	Only import	1,328 (12.0) [33.3]	811 (4.8) [20.3]	97 (5.0) [2.4]	1,261 (32.4) [31.6]	493 (10.6) [12.4]	3,990 (10.4) [100.0]
	Both export and import (two-way trader)	1,208 (10.9) [30.7]	104 (0.6) [2.6]	209 (10.8) [5.3]	229 (5.9) [5.8]	2,187 (47.1) [55.5]	3,937 (10.2) [100.0]
	Total	11,099 (100.0) [28.8]	16,989 (100.0) [44.1]	1,929 (100.0) [5.0]	3,890 (100.0) [10.1]	4,643 (100.0) [12.0]	38,550 (100.0) [100.0]

Reported are the number of cases, the column percentages in parenthesis ( ), and the row percentages in brackets [ ]. Only enterprises with one or more employees liable for paying social insurance and a turnover higher than epsilon17,081 in 2001 prices are considered. Tax groups and enterprises with a foreign legal form are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005.

TABLE 3

COMPARISON OF INTERNATIONALLY AVTIVE AND NON-ACTIVE MANUFACTURING ENTERPRISES IN WEST AND EAST GERMANY IN 2005

	Labour pr	oductivity	Number of employees liable to pay social insurance				
	Mean Index (in € 1,000) (in%)		Mean	Index (in %)			
West Germany							
All enterprises	143.1	100.0	25.5	100.0			
Enterprises that							
neither export nor import	110.7	85.8	6.2	41.7			
only export	141.0	98.4	11.1	59.6			
only import	145.5	101.5	12.7	77.6			
both export and import	199.8	125.9	72.0	232.2			
	East Ger	many		_			
All enterprises	91.4	100.0	16.3	100.0			
Enterprises that							
neither export nor import	76.6	90.5	6.9	53.5			
only export	99.7	99.2	14.4	96.0			
only import	105.1	109.3	15.2	107.5			
both export and import	131.1	127.9	52.9	268.8			

The index is computed as the percentage difference of the respective variable in an enterprise compared to the average value of all enterprises from the same 3-digit industry. Only enterprises with one or more employees liable for paying social insurance and a turnover higher than €17,081 in 2001 prices are considered. Tax groups, enterprises with a foreign legal form and the 1st and the 99th percentiles of the labour productivity distribution are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005.

TABLE 4

EXPORT AND IMPORT PRODUCTIVITY PREMIA IN MANUFACTURING ENTERPRISES IN WEST AND EAST GERMANY (2001-2005)

	Estimation of the log labour productivity in t							
		West G	ermany		East Germany			
	pooled regression		fixed effects*		pooled regression		fixed effects	
	1	2	1	2	1	2	1	2
Exporter dummy <sub>t</sub>	-	36.7 (0.00)	-	5.25 (0.00)	-	26.8 (0.00)	-	5.7 (0.00)
Only export dummy <sub>t</sub> (ß1)	18.9 (0.00)	-	4.17 (0.00)	-	12.3 (0.00)	-	4.6 (0.00)	-
Only import dummy <sub>t</sub> (ß2)	22.3 (0.00)	-	2.31 (0.00)	-	25.6 (0.00)	-	3.8 (0.00)	-
Two-way trader dummy $_{\rm t}$ (ß3)	55.8 (0.00)	-	8.79 (0.00)	-	47.8 (0.00)	-	10.4 (0.00)	-
Number of observations	652,219		195,623		141,299		141,299	

Reported are the estimated regression coefficients and the p-values (in parentheses) from two estimations of the log labour productivity at t. Model 1 contains an only export, an only import and a two-way trader dummy. ß1 is the average percentage productivity difference between exporters and non-exporters among enterprises that do not import. ß2 is the average percentage difference between importers and non-importers among non-exporters. ß3 is the average percentage difference between importer-exporters and enterprises that do neither export nor import. Model 2 contains an exporter dummy that shows the average percentage productivity difference between exporters and non-exporters. To facilitate the interpretation, the estimated coefficients of the dummy variables are transformed by 100(exp(ß)-1). Both models include the number of employees and its squared value, and a full set of interaction terms of year dummy variables and dummy variables for 3-digit level industries. Only enterprises with one or more employees liable for paying social insurance and a turnover higher than €17,081 in 2001 prices are considered. Tax groups, enterprises with a foreign legal form and from the 1st and the 99th percentiles of the labour productivity distribution are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005.

(\*) Due to limitations concerning the size of main memory available on the computers in the research data centre, it was not possible to estimate the fixed effects model with all West German enterprises. Therefore, the mean number of observations, the mean coefficients, and the mean p-values of five 30% random samples are reported. A documentation of the results for the five samples can be found in Table A.1 in the appendix.

TABLE 5A

IMPORT STARTERS VS. NON-STARTERS IN WEST AND EAST GERMANY IN 2005

		ductivity in 02	Number of liable to p insurance	Number of cases		
	Mean (in €1,000)	Index (in%)	Mean	Index (in %)	01 00303	
	West	Germany				
Non trading enterprises that start to import in 2005	112.0	103.8	9.6	149.3	2,122	
Enterprises that neither export nor import between 2002 and 2005	102.6	99.8	6.2	97.7	44,566	
Exporters that start to import in 2005	151.9	103.1	15.2	118.2	666	
Enterprises that only export between 2002 and 2005	150.2	99.4	13.2	96.7	3,702	
	East (	Germany			_	
Non trading enterprises that start to import in 2005	88.5	111.4	10.0	155.0	606	
Enterprises that neither export nor import between 2002 and 2005	71.2	99.5	6.7	97.4	12,614	
Exporters that start to import in 2005	112.9	105.8	21.2	112.2	87	
Enterprises that only export between 2002 and 2005	94.5	98.7	15.0	97.2	380	

The index is computed as the percentage difference of the respective variable in an enterprise compared to the average value of all enterprises from the same 3-digit industry. Only enterprises with one or more employees liable for paying social insurance and a turnover higher than €17,081 in 2001 prices are considered. Tax groups, enterprises with a foreign legal form and the 1st and the 99th percentiles of the labour productivity distribution are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005.

TABLE 5B

IMPORT STARTERS VS. NON-STARTERS IN WEST AND EAST GERMANY IN 2004

	Labour pro	oductivity in 01	Number of liable to p insurance	Number of cases		
	Mean (in €1,000)	Index (in%)	Mean	Index (in %)		
	West	Germany			_	
Non trading enterprises that start to import in 2004	118.9	106.5	8.5	140.9	2,033	
Enterprises that neither export nor import between 2001 and 2004	106.7	99.7	6.4	98.2	46,932	
Exporters that start to import in 2004	154.1	101.6	17.0	125.4	673	
Enterprises that only export between 2001and 2004	152.3	99.7	13.6	95.7	3,945	
East Germany						
Non trading enterprises that start to import in 2004	85.6	108.8	11.9	165.7	629	
Enterprises that neither export nor import between 2001 and 2004	73.6	99.6	6.9	96.9	13,483	
Exporters that start to import in 2004	116.5	106.0	17.1	113.0	65	
Enterprises that only export between 2001 and 2004	97.4	99.0	15.6	97.8	393	

The index is computed as the percentage difference of the respective variable in an enterprise compared to the average value of all enterprises from the same 3-digit industry. Only enterprises with one or more employees liable for paying social insurance and a turnover higher than €17,081 in 2001 prices are considered. Tax groups, enterprises with a foreign legal form and the 1st and the 99th percentiles of the labour productivity distribution are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005.

TABLE 6

SELF-SELECTION INTO IMPORT MARKETS OF MANUFACTURING ENTERPRISES IN WEST AND EAST GERMANY

		OLS estimation of the log labour productivity in t-3							
		West G	ermany			East Germany			
	t = 2	2004	t = 2005		t = 2004		t = 2005		
	1	2	1	2	1	2	1	2	
Non-trader that starts to import in t (dummy)	10.4 (0.00)	-	7.3 (0.00)	-	14.4 (0.00)	-	14.1 (0.00)	-	
Exporter that starts to import in t (dummy)	-	4.7 (0.06)	-	8.4 (0.00)	-	9.3 (0.30)	-	11.0 (0.20)	
Number of observations	48,965	4,618	46,688	4,368	14,112	458	13,220	467	

Reported are the estimated regression coefficients and the p-values (in parentheses) from the OLS estimation of the log labour productivity at t-3. To facilitate the interpretation, the estimated coefficient for the export dummy has been transformed by 100(exp(ß)-1). In model 1 the transformed coefficient shows the average percentage productivity difference at t-3 between import starters at t and enterprises with no international activities over the whole period (year t-3 to t). In model 2 the transformed coefficient shows the average percentage productivity difference at t-3 between exporters that start to import at t and exporters that do not start to import. Both models include the number of employees and its squared value, and a full set of dummy variables for 3-digit level industries. Only enterprises with one or more employees liable for paying social insurance and a turnover higher than €17,081 in 2001 prices are considered. Tax groups, enterprises with a foreign legal form and the 1st and the 99th percentiles of the labour productivity distribution are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005.

TABLE 7

GROWTH RATES OF IMPORT STARTERS AND NON-STARTERS IN WEST AND EAST GERMANY

	Growth rates of labour productivity between 2004 and 2005  Mean Index (in %) (in%)		Growth employees social in between 200	Number of cases	
			Mean (in %)	Index (in %)	
		(in%) Germany	(111 70)		
Non trading enterprises that start to import in 2003	4.4	100.7	2.0	101.2	607
Enterprises that neither export nor import between 2001 and 2005	3.9	99.9	0.4	100.0	36,255
Exporters that start to import in 2003	4.7	101.2	1.7	101.2	385
Enterprises that only export between 2001 and 2005	3.1 99.8		-0,2	99.8	2,757
	East (	Germany			
Non trading enterprises that start to import in 2003	7.3	98.5	1.9	102.2	116
Enterprises that neither export nor import between 2001 and 2005	7.4	100.0	-0.5	100.0	9,690
Exporters that start to import in 2003	8.7	101.4	7.3	105.5	49
Enterprises that only export between 2001 and 2005	2.4	99.7	1.5	99.0	266

The index is computed as the percentage difference of the respective growth rate in an enterprise compared to the average growth rate of all enterprises from the same 3-digit industry. Only enterprises with one or more employees liable for paying social insurance and a turnover higher than €17,081 in 2001 prices are considered. Tax groups, enterprises with a foreign legal form and the 1st and the 99th percentiles of the labour productivity distribution are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005.

TABLE 8

LEARNING-BY-IMPORTING IN MANUFACTURING ENTERPRISES IN WEST AND EAST GERMANY

	OLS estimation of growth of labour productivity (log labour productivity <sub>t+2</sub> – log labour productivity <sub>t+1)</sub>						
	West G	Sermany	East G	East Germany			
	1	2	1	2			
	t = 2003						
Non trading enterprise that starts to import in t (dummy)	1.2 (0.33)		-2.8 (0.40)				
Exporter that starts to import in t (dummy)		1.9 (0.18)		3.0 (0.51)			
Number of observations	36,862	3,142	9,806	315			

Reported are the estimated regression coefficients and the p-values (in parentheses) from the OLS estimation of log labour productivity in 2005 minus log labour productivity in 2004. To facilitate interpretation the estimated coefficient for the starter-dummy has been transformed by 100(exp(ß)-1). In model 1 the transformed coefficient shows the average productivity growth premium of import starters in 2003 compared to enterprises with no international activities two years after starting to import. In model 2 the transformed coefficient shows the average productivity growth premium of exporters that start to import in 2003 compared to enterprises that only export over the whole period two years after starting to import. Both models include the number of employees and its squared value plus a full set of 3-digit industry dummy variables. Only enterprises with one or more employees liable for paying social insurance and a turnover higher than €17,081 in 2001 prices are considered. Tax groups, enterprises with a foreign legal form and the 1st and the 99th percentiles of the labour productivity growth rate distribution are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005.

TABLE 9

GROWTH RATES OF MATCHED IMPORT STARTERS AND NON-STARTERS IN WEST AND EAST GERMANY

	Growth of labour productivity between 2004 and 2005					
	mean (in %)	Δ   = = = = = = = = = = = = = = = = = =		Observations (treated)		
	West Ger	many				
Non trading enterprises that start to import in 2003	4.6	1.8	0.449			
Matched Enterprises that neither export nor import between 2001 and 2005	2.7			517		
Exporters that start to import in 2003	4.7	3.1	0.224	343		
Matched Enterprises that only export between 2001 and 2005	1.6			343		
	East Ger	many		_		
Non trading enterprises that start to import in 2003	5.3	-15.8	0.047			
Matched Enterprises that neither export nor import between 2001 and 2005	21.1			102		
Exporters that start to import in 2003	13.0	12.2	0.096	35		
Matched Enterprises that only export between 2001 and 2005	0.7			00		

Reported are the mean labour productivity growth rates of the treated and the matched control groups, the average treatment effects (ATT) as well as the bootstrapped (1000 replications) p-values that indicates the statistical significance of the ATT. Matching was done by nearest neighbours propensity score matching. The propensity score was estimated from a probit regression of the import starter dummies on the log of labour productivity, number of employees, and a set of 3-digit industry dummy variables (all measured in 2002) plus the rate of growth of labour productivity between 2001 and 2002. The common support condition was imposed by dropping import starters (treated observations) whose propensity score was higher than the maximum or lower than the minimum propensity score of the control group. Only enterprises with one or more employees liable for paying social insurance and a turnover higher than €17,081 in 2001 prices are considered. Tax groups, enterprises with a foreign legal form and the 1st and the 99th percentiles of the labour productivity growth rate distribution are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005. The matching was done using Stata 10 and the psmatch2 command (version 3.0.0), see Leuven and Sianesi (2003).

TABLE A.1

EXPORT AND IMPORT PRODUCTIVITY PREMIA IN MANUFACTURING ENTERPRISES IN WEST GERMANY (2001-2005) – RESULTS FOR THE FIVE 30% RANDOM SAMPLES

		Estimation of the log labour productivity in t Model with fixed effects, West Germany								
	Random sample 1		Random sample 2		Random sample 3		Random sample 4		Random sample 5	
	1	2	1	2	1	2	1	2	1	2
Exporter dummy <sub>t</sub>	-	4.98 (0.00)	-	5.00 (0.00)	-	5.23 (0.00)		5.79 (0.00)	-	5.25 (0.00)
Only export dummy <sub>t</sub> (ß1)	4.12 (0.00)	-	3.81 (0.00)	-	4.14 (0.00)	-	4.28 (0.00)	-	4.51 (0.00)	-
Only import dummy <sub>t</sub> (ß2)	2.71 (0.00)	-	2.81 (0.00)	-	2.20 (0.00)	-	2.02 (0.00)	-	1.83 (0.00)	-
Two-way trader dummy <sub>t</sub> (ß3)	8.60 (0.00)	-	9.12 (0.00)	-	8.73 (0.00)	-	9.57 (0.00)	-	7.90 (0.00)	-
Number of observations	195	,591	195	,633	195	,777	194	,562	196	,553

Due to limitations concerning the size of the main memory available on the computers in the research data centre, it was not possible to estimate the fixed effects model with all West German enterprises. Therefore, the mean values of the estimated regression coefficients and the p-values of five 30% random samples taken from all firms are reported in Table 4. This table reports the results for each of these samples. Model 1 contains an only export, an only import and a two-way trader dummy. ß1 is the average percentage productivity difference between exporters and non-exporters among enterprises that do not import. B2 is the average percentage difference between importers and non-importers among non-exporters. B3 is the average percentage difference between importer-exporters and enterprises that do neither export nor import. Model 2 contains an exporter dummy that shows the average percentage productivity difference between exporters and non-exporters. To facilitate the interpretation, the estimated coefficients of the dummy variables are transformed by 100(exp(ß)-1). Both models include the number of employees and its squared value, and a full set of interaction terms of year dummy variables and dummy variables for 3-digit level industries. Only enterprises with one or more employees liable for paying social insurance and a turnover higher than €17,081 in 2001 prices are considered. Tax groups, enterprises with a foreign legal form and the 1st and the 99th percentiles of the labour productivity distribution are excluded from all computations. Data source: German turnover tax statistics panel 2001-2005.

# **Working Paper Series in Economics**

(see www.leuphana.de/vwl/papers for a complete list)

No.88:

the field Firm Level Data, June 2008

No.105:	Markus Groth: Kosteneffizienter und effektiver Biodiversitätsschutz durch Ausschreibungen und eine ergebnisorientierte Honorierung: Das Modellprojekt "Blühendes Steinburg". November 2008
No.104:	Alexander Vogel & Joachim Wagner: Export, Import und Produktivität wissensintensiver KMUs in Deutschland. Oktober 2008
No.103:	Christiane Clemens & Maik Heinemann: On Entrepreneurial Risk – Taking and the Macroeconomic Effects Of Financial Constraints, October 2008
No.102:	Helmut Fryges & Joachim Wagner: Exports and Profitability – First Evidence for German Manufacturing Firms. October 2008
No.101:	Heike Wetzel: Productivity Growth in European Railways: Technological Progress, Efficiency Change and Scale Effects. October 2008
No.100:	Henry Sabrowski: Inflation Expectation Formation of German Consumers: Rational or Adaptive? October 2008
No.99:	Joachim Wagner: Produktdifferenzierung in deutschen Industrieunternehmen 1995 – 2004: Ausmaß und Bestimmungsgründe, Oktober 2008
No.98:	Jan Kranich: Agglomeration, vertical specialization, and the strength of industrial linkages, September 2008
No.97:	Joachim Wagner: Exports and firm characteristics - First evidence from Fractional Probit Panel Estimates, August 2008
No.96:	Nils Braakmann: The smoking wage penalty in the United Kingdom: Regression and matching evidence from the British Household Panel Survey, August 2008
No.95:	Joachim Wagner: Exportaktivitäten und Rendite in niedersächsischen Industrieunternehmen, August 2008
No.94:	[publiziert in: Statistische Monatshefte Niedersachsen 62 (2008), 10,552-560]  Joachim Wagner: Wirken sich Exportaktivitäten positiv auf die Rendite von deutschen Industrieunternehmen aus?, August 2008  [publiziert in: Wirtschaftsdienst, 88 (2008) 10, 690-696]
No.93:	Claus Schnabel & Joachim Wagner: The aging of the unions in West Germany, 1980-2006, August 2008 [forthcoming in: Jahrbücher für Nationalökonomie und Statistik]
No.92:	Alexander Vogel and Stefan Dittrich: The German turnover tax statistics panels, August 2008 [forthcoming in: Schmollers Jahrbuch 128 (2008)]
No.91:	Nils Braakmann: Crime does pay (at least when it's violent!) – On the compensating wage differentials of high regional crime levels, July 2008
No.90:	Nils Braakmann: Fields of training, plant characteristics and the gender wage gap in entry wages among skilled workers – Evidence from German administrative data, July 2008
No.89:	Alexander Vogel: Exports productivity in the German business services sector: First evidence from the Turnover Tax Statistics panel, July 2008

Joachim Wagner: Improvements and future challenges for the research infrastructure in

- No.87: *Markus Groth:* A review of the German mandatory deposit for one-way drinks packaging and drinks packaging taxes in Europe, June 2008
- No.86: *Heike Wetzel:* European railway deregulation. The influence of regulatory ans environmental conditions on efficiency, May 2008
- No.85: Nils Braakmann: Non scholae, sed vitae discimus! The importance of fields of study for the gender wage gap among German university graduates during market entry and the first years of their careers, May 2008
- No.84: *Markus Groth:* Private ex-ante transaction costs for repeated biodiversity conservation auctions: A case study, May 2008
- No.83: Jan Kranich: R&D and the agglomeration of industries, April 2008
- No.82: Alexander Vogel: Zur Exporttätigkeit unternehmensnaher Dienstleister in Niedersachsen Erste Ergebnisse zu Export und Produktivität auf Basis des Umsatzsteuerstatistikpanels, April 2008
- No.81: *Joachim Wagner:* Exporte und Firmenerfolg: Welche Firmen profitieren wie vom internationalen Handel?, März 2008
- No.80: Stefan Baumgärtner: Managing increasing environmental risks through agro-biodiversity and agri-environmental policies, March 2008
- No.79: *Thomas Huth:* Die Quantitätstheorie des Geldes Eine keynesianische Reformulierung, März 2008
- No.78: *Markus Groth:* An empirical examination of repeated auctions for biodiversity conservation contracts, March 2008
- No.77: *Nils Braakmann:* Intra-firm wage inequality and firm performance First evidence from German linked employer-employee-data, February 2008
- No.76: *Markus Groth*: Perspektiven der Nutzung von Methanhydraten als Energieträger Eine Bestandsaufnahme, Februar 2008
- No.75: Stefan Baumgärtner, Christian Becker, Karin Frank, Birgit Müller & Christian Quaas:
  Relating the philosophy and practice of ecological economics. The role of concepts,
  models, and case studies in inter- and transdisciplinary sustainability research, January
  2008
  [publisched in: Ecological Economics 67 (2008), 3, 384-393]
- No.74: Thorsten Schank, Claus Schnabel & Joachim Wagner: Higher wages in exporting firms: Self-selection, export effect, or both? First evidence from German linked employer-employee data, January 2008
- No.73: Institut für Volkswirtschaftslehre: Forschungsbericht 2007, Januar 2008
- No.72: Christian Growitsch and Heike Wetzel:Testing for economies of scope in European railways: An efficiency analysis, December 2007
  [revised version of Working Paper No. 29,
  forthcoming in: Journal of Transport Economics and Policy]
- No.71: Joachim Wagner, Lena Koller and Claus Schnabel: Sind mittelständische Betriebe der Jobmotor der deutschen Wirtschaft?, Dezember 2007 [publiziert in: Wirtschftsdienst 88 (2008), 2, 130-135]
- No.70: *Nils Braakmann:* Islamistic terror, the war on Iraq and the job prospects of Arab men in Britain: Does a country's direct involvement matter?, December 2007

- No.69: *Maik Heinemann:* E-stability and stability learning in models with asymmetric information,

  December 2007
- No.68: *Joachim Wagner:* Exporte und Produktivität in Industriebetrieben Niedersachsen im interregionalen und internationalen Vergleich, Dezember 2007
- No.67: Stefan Baumgärtner and Martin F. Quaas: Ecological-economic viability as a criterion of strong sustainability under uncertainty, November 2007
- No.66: *Kathrin Michael:* Überbrückungsgeld und Existenzgründungszuschuss Ergebnisse einer schriftlichen Befragung drei Jahre nach Gründungsbeginn, November 2007
- No.65: The International Study Group on Export and Productivity: Exports and Productivity –
  Comparable Evidence for 14 Countries, November 2007
  [forthcoming in: Review of World Economics 144 (2008), 4]
- No.64: Lena Koller, Claus Schnabel und Joachim Wagner: Freistellung von Betriebsräten Eine Beschäftigungsbremse?, November 2007
  [publiziert in: Zeitschrift für Arbeitsmarktforschung, 41 (2008), 2/3, 305-326]
- No.63: Anne-Kathrin Last: The Monetary Value of Cultural Goods: A Contingent Valuation Study of the Municipal Supply of Cultural Goods in Lueneburg, Germany, October 2007
- No.62: Thomas Wein und Heike Wetzel: The Difficulty to Behave as a (regulated) Natural Monopolist The Dynamics of Electricity Network Access Charges in Germany 2002 to 2005, September 2007
- No.61: Stefan Baumgärtner und Martin F. Quaas: Agro-biodiversity as natural insurance and the development of financial insurance markets, September 2007 [published in: A. Kontoleon, U. Pascual and M. Smale (eds.): Agrobiodiversity, conservation and economic development, Routledge, London, 293-317]
- No.60: Stefan Bender, Joachim Wagner, Markus Zwick: KombiFiD Kombinierte Firmendaten für Deutschland, September 2007
- No.59: Jan Kranich: Too much R&D? Vertical differentiation in a model of monopolistic competition, August 2007
- No.58: Christian Papilloud und Ingrid Ott: Convergence or mediation? Experts of vulnerability and the vulnerability of experts' discourses on nanotechnologies a case study, July 2007
- [published in: European Journal of Social Science Research 21 (2008), 1, 41-64]

  No.57: Ingrid Ott und Susanne Soretz: Governmental activity, integration and agglomeration,
- July 2007

  [published in: ICFAI Journal of Managerial Economics 5 (2008), 2, 28-47]
- No.56: Nils Braakmann: Struktur und Erfolg von Ich-AG-Gründungen: Ergebnisse einer Umfrage im Arbeitsagenturbezirk Lüneburg, Juli 2007
  [revidierte Fassung erscheint in: Richter, J., Schöning, S. & Wetzel, H., Mittelstand 2008.
  - Aktuelle Forschungsbeiträge zu gesellschaftlichen und finanzwirtschaftlichen Herausforderungen, Frankfurt am Main: Peter Lang, 2008]
- No.55: *Nils Braakmann:* Differences in the earnings distribution of self- and dependent employed German men evidence from a quantile regression decomposition analysis, July 2007
- No.54: *Joachim Waagner:* Export entry, export exit, and productivity in German Manufacturing Industries, June 2007

  [published in: International Journal of the Economics of Business 15 (2008), 2, 169-180]

- No.53: Nils Braakmann: Wirkungen der Beschäftigungspflicht schwerbehinderter Arbeitnehmer Erkenntnisse aus der Einführung des "Gesetzes zur Bekämpfung der Arbeitslosigkeit Schwerbehinderter", Juni 2007 [revidierte Fassung erscheint in: Zeitschrift für Arbeitsmarktforschung/ Journal for Labour Market Research 41 (2008),1, 9-24]
- No.52: Jan Kranich und Ingrid Ott: Regionale Spitzentechnologie auf internationalen Märkten,
  Juni 2007
  [erscheint in: Merz, J. und Schulte, R. (Hrsg.): Neue Ansätze der MittelstandsForschung,
  Münster, 2007]
- No.51: Joachim Wagner: Die Forschungspotenziale der Betriebspaneldaten des Monatsberichts im Verarbeitenden Gewerbe, Mai 2007 [publiziert in: AStA Wirtschafts- und Sozialwirtschaftliches Archiv 2 (2008), 3, 209-221]
- No.50: Stefan Baumgärtner, Frank Jöst und Ralph Winkler: Optimal dynamic scale and structure of a multi-pollution economy, May 2007
  [forthcoming in: Ecological Economics]
- No.49: Helmut Fryges und Joachim Wagner: Exports and productivity growth First evidence from a continuous treatment approach, May 2007
  [forthcoming in: Review of World Economics]
- No.48: *Ulrich Kaiser und Joachim Wagner:* Neue Möglichkeiten zur Nutzung vertraulicher amtlicher Personen- und Firmendaten, April 2007

  [publiziert in: Perspektiven der Wirtschaftspolitik 9 (2008), 3, 329-349]
- No.47: Joachim Wagner: Jobmotor Mittelstand? Arbeitsplatzdynamik und Betriebsgröße in der westdeutschen Industrie, April 2007 [publiziert in: Vierteljahrshefte zur Wirtschaftsforschung, 76 (2007), 3, 76-87]
- No.46: Christiane Clemens und Maik Heinemann: Credit Constraints, Idiosyncratic Risks, and the Wealth Distribution in a Heterogenous Agent Model, March 2007
- No.45: *Jan Kranich:* Biotechnologie und Internationalisierung. Ergebnisse der Online-Befragung, März 2007
- No.44: Joachim Wagner: Entry, exit and productivity. Empirical results for German manufacturing industries, March 2007
  [forthcoming in: German Economic Review]
- No.43: *Joachim Wagner:* Productivity and Size of the Export Market Evidence for West and East German Plants, 2004, March 2007

  [publiziert in: Jahrbücher für Nationalökonomie und Statistik, 227 (2007), 4, 403-408]
- No.42: Joachim Wagner: Why more West than East German firms export, March 2007 [forthcoming in: International Economics and Economic Policy]
- No.41: *Joachim Wagner:* Exports and Productivity in Germany, March 2007 [publiziert in: Applied Economics Quarterly 53 (2007), 4, 353-373]
- No.40: Lena Koller, Klaus Schnabel und Joachim Wagner: Schwellenwerte im Arbeitsrecht.

  Höhere Transparenz und Effizienz durch Vereinheitlichung, Februar 2007

  [publiziert in: Perspektiven der Wirtschaftspolitik, 8 (2007), 3, 242-255]
- No.39: *Thomas Wein und Wiebke B. Röber:* Sind ausbildende Handwerksbetriebe erfolgreicher?, Januar 2007
- No.38: Institut für Volkswirtschaft: Forschungsbericht 2006, Januar 2007

- No.37: Nils Braakmann: The impact of September 11<sup>th</sup>, 2001 on the job prospects of foreigners with Arab background Evidence from German labor market data, January 2007 [revised version forthcoming as "The impact of September 11th, 2001 on the employment prospects of Arabs and Muslims in the German labor market" in Jahrbücher für Nationalökonomie und Statistik / Journal of Economics and Statistics]
- No.36: *Jens Korunig:* Regulierung des Netzmonopolisten durch Peak-load Pricing?, Dezember 2006
- No.35: Nils Braakmann: Die Einführung der fachkundigen Stellungnahme bei der Ich-AG,
  November 2006
  [erscheint in: Schulte, Reinhard: Neue Ansätze der MittelstandsForschung, Münster etc.:
  Lit, 2008]
- No.34: *Martin F. Quaas and Stefan Baumgärtner:* Natural vs. financial insurance in the management of public-good ecosystems, October 2006 [published in: Ecological Economics 65 (2008), 2, 397-406]
- No.33: Stefan Baumgärtner and Martin F. Quaas: The Private and Public Insurance Value of Conservative Biodiversity Management, October 2006
- No.32: *Ingrid Ott and Christian Papilloud:* Converging institutions. Shaping the relationships between nanotechnologies, economy and society, October 2006 [published in: Bulletin of Science, Technology & Society 2007 (27), 4, 455-466]
- No.31: Claus Schnabel and Joachim Wagner: The persistent decline in unionization in western and eastern Germany, 1980-2004: What can we learn from a decomposition analysis?, October 2006

  [published in: Industrielle Beziehungen/The German Journal of Industrial Relations 14 (2007), 118-132]
- No.30: Ingrid Ott and Susanne Soretz: Regional growth strategies: fiscal versus institutional governmental policies, September 2006
  [published in: Economic Modelling 25 (1008), 605-622]
- No.29: Christian Growitsch and Heike Wetzel: Economies of Scope in European Railways: An Efficiency Analysis, July 2006
- No.28: Thorsten Schank, Claus Schnabel and Joachim Wagner: Do exporters really pay higher wages? First evidence from German linked employer-employee data, June 2006 [published in in: Journal of International Economics 72 (2007), 1, 52-74]
- No.27: Joachim Wagner: Markteintritte, Marktaustritte und Produktivität
  Empirische Befunde zur Dynamik in der Industrie, März 2006
  [publiziert in: AStA Wirtschafts- und Sozialwirtschaftliches Archiv 1 (2007), 3, 193-203]
- No.26: Ingrid Ott and Susanne Soretz: Governmental activity and private capital adjustment,

  March 2006

  [forthcoming in: Icfai Journal of Managerial Economics]
- No.25: Joachim Wagner: International Firm Activities and Innovation:

  Evidence from Knowledge Production Functions for German Firms, March 2006

  [published in: The Icfai Journal of Knowledge Management VI (2008), 2, 47-62]

- No.24: Ingrid Ott und Susanne Soretz: Nachhaltige Entwicklung durch endogene
  Umweltwahrnehmung, März 2006
  publiziert in: Clemens, C., Heinemann, M. & Soretz, S., Auf allen Märkten zu Hause
  (Gedenkschrift für Franz Haslinger), Marburg: Metropolis, 2006, 233-256
- No.23: John T. Addison, Claus Schnabel, and Joachim Wagner: The (Parlous) State of German Unions, February 2006
  [published in: Journal of Labor Research 28 (2007), 3-18]
- No.22: Joachim Wagner, Thorsten Schank, Claus Schnabel, and John T. Addison: Works
  Councils, Labor Productivity and Plant Heterogeneity: First Evidence from Quantile
  Regressions, February 2006
  [published in: Jahrbücher für Nationalökonomie und Statistik 226 (2006), 505 518]
- No.21: Corinna Bunk: Betriebliche Mitbestimmung vier Jahre nach der Reform des BetrVG: Ergebnisse der 2. Befragung der Mitglieder des Arbeitgeberverbandes Lüneburg Nordostniedersachsen, Februar 2006
- No.20: Jan Kranich: The Strength of Vertical Linkages, July 2006
- No.19: Jan Kranich und Ingrid Ott: Geographische Restrukturierung internationaler
  Wertschöpfungsketten Standortentscheidungen von KMU aus regionalökonomischer
  Perspektive, Februar 2006
  [publiziert in: Merz, J. und Schulte, R. (Hrsg.): Fortschritte in der MittelstandsForschung,
  Münster, 2006, 113-129]
- No.18: *Thomas Wein und Wiebke B. Röber:* Handwerksreform 2004 Rückwirkungen auf das Ausbildungsverhalten Lüneburger Handwerksbetriebe?, Februar 2006
- No.17: Wiebke B. Röber und Thomas Wein: Mehr Wettbewerb im Handwerk durch die Handwerksreform?, Februar 2006
- No.16: Joachim Wagner: Politikrelevante Folgerungen aus Analysen mit wirtschaftsstatistischen Einzeldaten der Amtlichen Statistik, Februar 2006 [publiziert in: Schmollers Jahrbuch 126 (2006) 359-374]
- No.15: Joachim Wagner: Firmenalter und Firmenperformance
  Empirische Befunde zu Unterschieden zwischen jungen und alten Firmen
  in Deutschland, September 2005
  [publiziert in: Lutz Bellmann und Joachim Wagner (Hrsg.), Betriebsdemographie
  (Beiträge zur Arbeitsmarkt- und Berufsforschung, Band 305), Nürnberg: IAB der BA,
  83-111]
- No.14: *Joachim Wagner:* German Works Councils and Productivity: First Evidence from a Nonparametric Test, September 2005 [published in: Applied Economics Letters 115 (2008), 727-730]
- No.13: Lena Koller, Claus Schnabel und Joachim Wagner: Arbeitsrechtliche Schwellenwerte und betriebliche Arbeitsplatzdynamik: Eine empirische Untersuchung am Beispiel des Schwerbehindertengesetzes, August 2005
  [publiziert in: Zeitschrift für ArbeitsmarktForschung/ Journal for Labour Market Research 39 (2006), 181-199]

- No.12: Claus Schnabel and Joachim Wagner: Who are the workers who never joined a union? Empirical evidence from Germany, July 2005
  [published in: Industrielle Beziehungen/ The German Journal of Industrial Relations 13 (2006), 118-131]
- No.11: Joachim Wagner: Exporte und Produktivität in mittelständischen Betrieben
  Befunde aus der niedersächsischen Industrie (1995 2004), June 2005
  [publiziert in: Niedersächsisches Landesamt für Statistik, Statistische Berichte
  Niedersachsen, Sonderausgabe: Tagung der NLS am 9. März 2006, Globalisierung und
  regionale Wirtschaftsentwicklung Datenlage und Datenbedarf in Niedersachsen.
  Hannover, Niedersächsisches Landesamt für Statistik, Juli 2006, 18 29]
- No.10: Joachim Wagner: Der Noth gehorchend, nicht dem eignen Trieb.

  Nascent Necessity and Opportunity Entrepreneurs in Germany.

  Evidence from the Regional Entrepreneurship Monitor (REM), May 2005

  [published in: RWI: Mitteilungen. Quarterly 54/ 55 (2003/04), 287-303

  {published June 2006}]
- No. 9: Gabriel Desgranges and Maik Heinemann: Strongly Rational Expectations Equilibria with Endogenous Acquisition of Information, March 2005
- No. 8: Joachim Wagner: Exports, Foreign Direct Investment, and Productivity: Evidence from German Firm Level Data, March 2005
  [published in: Applied Economics Letters 13 (2006), 347-349]
- No. 7: *Thomas Wein:* Associations' Agreement and the Interest of the Network Suppliers The Strategic Use of Structural Features, March 2005
- No. 6: Christiane Clemens and Maik Heinemann: On the Effects of Redistribution on Growth and Entrepreneurial Risk-Taking, March 2005
- No. 5: Christiane Clemens and Maik Heinemann: Endogenous Redistributive Cycles An overlapping Generations Approach to Social Conflict and Cyclical Growth, March 2005
- No. 4: Joachim Wagner: Exports and Productivity: A Survey of the Evidence from Firm Level
  Data, March 2005
  [published in: The World Economy 30 (2007), 1, 60-82]
- No. 3: Thomas Wein and Reimund Schwarze: Is the Market Classification of Risk Always Efficient? Evidence from German Third Party Motor Insurance, March 2005
- No. 2: Ingrid Ott and Stephen J. Turnovsky: Excludable and Non-Excludable Public Inputs: Consequences for Economic Growth, June 2005 (Revised version) [published in: Economica 73 (2006), 292, 725-742 also published as CESifo Working Paper 1423]
- No. 1: Joachim Wagner: Nascent and Infant Entrepreneurs in Germany.

  Evidence from the Regional Entrepreneurship Monitor (REM), March 2005

  [erschienen in: Joachim Merz, Reinhard Schulte (Hrsg.), Neue Ansätze der MittelstandsForschung, Berlin: Lit Verlag 2008, S.395-411]

Leuphana Universität Lüneburg
Institut für Volkswirtschaftslehre
Postfach 2440
D-21314 Lüneburg

Tel.: ++49 4131 677 2321 email: brodt@leuphana.de www.leuphana.de/vwl/papers